Author’s response to reviews

Title: Social Network Properties and Self-Rated Health in Later Life Comparisons from the Korean Social Life, Health, and Aging Project and the National Social Life, Health and Aging Project

Authors:

Yoosik Youm (yoosik@yonsei.ac.kr)
Edward O Laumann (ob01@uchicago.edu)
Kenneth F Ferraro (ferraro@purdue.edu)
Linda J Waite (l-waite@uchicago.edu)
Hyeon Chang Kim (hckim@yuhs.ac)
Yeong-Ran Park (yeongranpark@daum.net)
Sang Hui Chu (shchu@yuhs.ac)
Won-tak Joo (yochicken@naver.com)
Jina Lee (lovejirosang@gmail.com)

Version: 2 Date: 30 April 2014

Author’s response to reviews: see over
Dear editorial board of Journal of Gerontologist:

Please find enclosed revised version manuscript: "Social Network Properties and Self-Rated Health in Later Life: Comparisons from the Korean Social Life, Health, and Aging Project and the National Social Life, Health and Aging Project", by Yoosik Youm and others to be submitted as an original design paper to the Journal of Gerontologist for consideration of publication.

As noted by editorial office, title page, competing interests, ethics and line numbering are added.

We apologize for the inconvenience.

Sincerely yours,

Yoosik Youm on behalf of the authors of the manuscript

Corresponding author: Yoosik Youm. Yonsei-ro 50. Sociology Department. Yonsei University, Seoul, South Korea. 120-749. yoosik@yonsei.ac.kr. Phone number: 82-2-2123-2431, fax number: 82-2-2123-2420.